Author's response to reviews

Title: Injury in China: A systematic review of injury surveillance studies conducted in Chinese hospital emergency departments

Authors:

Michael Fitzharris (Michael.Fitzharris@monash.edu)
James Yu (jyu@george.org.cn)
Naomi Hammond (nhammond@george.org.au)
Colman Taylor (ctaylor@george.org.au)
Yangfeng Wu (ywu@george.org.cn)
Simon Finfer (sfinner@george.org.au)
John Myburgh (jmyburgh@george.org.au)

Version: 5 Date: 23 October 2011

Author's response to reviews: see over
18 October 2011

To:
The BMC Series Editorial Production Team

Dear Team

RE: MS : 4255845854853337

Finalisation of manuscript layout: Injury in China: A systematic review of injury surveillance studies conducted in Chinese hospital emergency departments

The changes for the finalisation of the manuscript have been undertaken in accordance with your email dated, 14 October 2011. These are detailed below.

Thank-you for your support in the production of our manuscript.

Sincerely

Dr Michael Fitzharris, on behalf of the Co-Authors.

<table>
<thead>
<tr>
<th>PRODUCTION TEAM: Major revisions (we require the author to make these changes)</th>
</tr>
</thead>
<tbody>
<tr>
<td>References: Please make sure references are cited in numerical order.</td>
</tr>
<tr>
<td>All references have been corrected, and appear in text and in order.</td>
</tr>
<tr>
<td>Tables: Please ensure that tables are placed in both columns and rows. If</td>
</tr>
<tr>
<td>this is not possible then the information can be represented in the main</td>
</tr>
<tr>
<td>text as a list of bullet points but cannot be referred to as a table.</td>
</tr>
<tr>
<td>This was clarified and tables were seen to be ok (email from Ms Robyn Hughes,</td>
</tr>
<tr>
<td>21/10 following clarification being sought.</td>
</tr>
<tr>
<td>Additional files: Please give each additional file a separate title before</td>
</tr>
<tr>
<td>the description in the additional files section.</td>
</tr>
<tr>
<td>Done, added a title to each.</td>
</tr>
<tr>
<td>PRODUCTION TEAM</td>
</tr>
<tr>
<td>-----------------</td>
</tr>
<tr>
<td>Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements</td>
</tr>
<tr>
<td>Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: <a href="http://www.biomedcentral.com/info/ifora/figures">http://www.biomedcentral.com/info/ifora/figures</a></td>
</tr>
<tr>
<td>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance</td>
</tr>
</tbody>
</table>